Author’s response to reviews

Title: Retention in care and virological failure among adult HIV+ patients on second-line ART in Rwanda: A national representative study

Authors:

Sabin Nsanzimana (nsabinco@gmail.com)
Muhammed Semakula (semakulam@gmail.com)
Vedaste Ndahindwa (ndahindwa@gmail.com)
Eric Remera (ericremera@gmail.com)
Dieudonne Sebuhoro (sebudieu@yahoo.fr)
Jean Paul Uwizihiwe (uwizihiwejp@gmail.com)
Nathan Ford (nathanpford@gmail.com)
Marcel Tanner (marcel.tanner@unibas.ch)
Steve Kanters (steve.kanters@precisionhealtheconomics.com)
Edward Mills (emills@mteksciences.com)
Heiner Bucher (Heiner.Bucher@usb.ch)

Version: 4 Date: 18 Mar 2019

Author’s response to reviews:

Dear Editor,

I thank you for your feedback and opportunity to move this manuscript to another stage
I have considered all your suggestions:

1. Participants consent waiver added in the ethical considerations and declarations
2. We also clarified that Datasets used during the current study are available from the corresponding author on reasonable request.

3. Author's roles updated specifically for Dr Jean Paul Uwizihwe

4. We also added under the same section that all authors have read and approved the manuscript

The cleaned version of the manuscript is herewith uploaded accordingly